Dear Editors:
Thank you for your affirmation, we have re-adjusted according to your revised comments, as follows:
1. Bacterial name and species should be stated with italic font.
   already edited
2. There are some typos errors.
   already verified
3. The purpose of the study should be reported in the last paragraph of the introduction.
   already edited
4. L. monocytogenes how was identified? molecular or culture?
   Bilateral blood culture and cerebrospinal fluid NGS
5. The author should be prepare a table with complete feature characteristics, treatment course and clinical outcomes of similar previous published cases.
   already edited in table 1
6. The author should be cited relevant studies over last 3 years ago
   Studies that have been added for nearly 3 years
7. uniform presentation should be used for figures showing the same or similar contents; for example, “Figure 1 Pathological changes of atrophic gastritis after treatment. A: ...; B: ...; C: ...; D: ...; E: ...; F: ...; G: ...
   Image modified as requested
8. Before its final acceptance, the author(s) must provide the Signed Consent for Treatment Form(s) or Document(s)
   provided in the files
9. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed
   provided in the files
10. Please check and confirm whether the figures are original (i.e. generated de novo by the author(s) for this paper). If the picture is ‘original’, the author needs to add the following copyright information to the bottom right-hand side of the picture in PowerPoint (PPT): Copyright ©The Author(s) 2022.
   Have marked in ppt

We deeply appreciate your consideration of our manuscript. If you have any queries, please don’t hesitate to contact me at the address below.

Thank you and best regards.

Yours sincerely,
Shuangquan Yan